Author's response to reviews

Title: Tuscan black kale sprout extract bioactivated with myrosinase: a novel natural product for neuroprotection by inflammatory and oxidative response during cerebral ischemia/reperfusion injury in rat

Authors:

Sabrina Giacoppo (giacoppo.sabrina@hotmail.it)
Maria Galuppo (mariagaluppo@gmail.com)
Gina R De Nicola (gina.denicola@cracin.it)
Renato Iori (renato.iori48@gmail.com)
Placido Bramanti (bramanti.dino@gmail.com)
Emanuela Mazzon (emazzon.irccs@gmail.com)

Version: 2  
Date: 25 June 2015

Author's response to reviews: see over
Dear Editor,

We would like to submit the enclosed manuscript: “Tuscan black kale sprout extract bioactivated with myrosinase: a novel natural product for neuroprotection by inflammatory and oxidative response during cerebral ischemia/reperfusion injury in rat” to be considered as possible publication in BMC Complementary & Alternative Medicine.

The present study shows the possible therapeutic use of the bioactive phytochemical purified by Tuscan black kale sprout. In the present study, this natural product has been tested in order to assess its therapeutic effect into protect brain tissue by experimental cerebral ischemia/reperfusion.

This study was carried out in strict accordance with the recommendations in the guide for the care and use of laboratory animals of the National Institutes of Health. The protocol was approved by the Ministry of Health “General Direction of animal health and veterinary drug”.

The manuscript has never been published, or under the consideration for publication elsewhere.

We confirm that the manuscript has been read and approved by all named authors. We further confirm that the order of authors listed in the manuscript has been approved by all of us.

No conflict of interest is to be declared.

Also we suggest we name of potential reviewers for this manuscript:

1. Anika E. Wagner Institute of Human Nutrition and Food Science, University of Kiel, Hermann-Rodewald-Strasse 6, 24118 Kiel, Germany
   Email: wagner@molecularnutrition.uni-kiel.de
2. Maria Traka Food and Health Programme, Institute of Food Research, Norwich Research Park, Norwich, NR4 7UA, UK.
   Email: maria.traka@bbsrc.ac.uk

We strongly hope that this could be a good job, in accordance with the expectations of your journal and worthy to be published.

Sincerely yours,

THE AUTHOR FOR CORRESPONDENCE

Dr. Emanuela Mazzon
IRCCS - Centro Neurolesi “Bonino-Pulejo”, Messina (Italy)